Author's response to reviews

Title: Electrical impedance along connective tissue planes associated with acupuncture meridians

Authors:

Andrew C Ahn (aahn@hms.harvard.edu)
Junru Wu (Junru.Wu@uvm.edu)
Gary J Badger (Gary.Badger@uvm.edu)
Richard Hammerschlag (rhammerschlag@ocom.edu)
Helene M Langevin (Helene.Langevin@uvm.edu)

Version: 4 Date: 2 May 2005

Author's response to reviews: see over
May 2, 2005

The Editors
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UK

Subject: Final Manuscript

Re: Manuscript number 1906187103489521

Dear Editors:

Thank you for your response and your decision to accept our manuscript entitled: “Electrical Impedance along Connective Tissue Planes associated with Acupuncture Meridians”. We have formatted the manuscript to comply with the journal’s requirements. The changes are detailed here:

1. Section Headings. The title of the manuscript has been changed to sentence case (see Page 1). All other headings and sub-headings have also been verified as being in sentence case format.
2. Title page. The manuscript title, authors, authors’ affiliation, e-mail address, and corresponding author information are listed in the title page. All other information has been deleted.
3. Manuscript sections. The manuscript is ordered accordingly: Abstract; Background; Methods; Results; Discussion; Conclusions; Competing interests; Authors’ contributions; Acknowledgements; References; and Figure legends. No abbreviation, table legends, or description of additional files are needed for our manuscript.
4. Conclusion. The last paragraph serves as the Conclusion for this manuscript. (Please see page 14, last paragraph).

We have also thoroughly checked our manuscript for any errors. We hope this manuscript meets your approval. Thank you for your time.

Sincerely,

Andrew Ahn, MD MPH
Harvard Medical School
401 Park Drive Suite 22A-West
Boston, MA 02215
aahn@hms.harvard.edu
(W) 617-384-8581
(F) 617-384-8555